



Article Fiber-Optic Hydrophone Based on Michelson's Interferometer with Active Stabilization for Liquid Volume Measurement

Welton Sthel Duque 🗅, Camilo Arturo Rodríguez Díaz *🗅, Arnaldo Gomes Leal-Junior 🗅 and Anselmo Frizera 🗅

Telecommunications Laboratory (LABTEL), Graduate Program in Electrical Engineering, Federal University of Espírito Santo (UFES), Vitória 29075-910, ES, Brazil; wduque@gmail.com (W.S.D.); leal-junior.arnaldo@ieee.org (A.G.L.-J.); frizera@ieee.org (A.F.)

* Correspondence: c.rodriguez.2016@ieee.org

Abstract: Sensing technologies using optical fibers have been studied and applied since the 1970s in oil and gas, industrial, medical, aerospace, and civil areas. Detecting ultrasound acoustic waves through fiber-optic hydrophone (FOH) sensors can be one solution for continuous measurement of volumes inside production tanks used by these industries. This work presents an FOH system composed of two optical fiber coils made with commercial single mode fiber (SMF) working in the sensor head of a Michelson's interferometer (MI) supported by an active stabilization mechanism that drives another optical coil wound around a piezoelectric actuator (PZT) in the reference arm to mitigate external mechanical and thermal noise from the environment. A 1000 mL glass graduated cylinder filled with water is used as a test tank, inside which the sensor head and an ultrasound source are placed. For detection, amplitudes and phases are measured, and machine learning algorithms predict their respective liquid volumes. The acoustic waves create patterns electronically detected with resolution of 1 mL and sensitivity of 340 mrad/mL and 70 mvolts/mL. The nonlinear behavior of both measurands requires classification, distance metrics, and regression algorithms to define an adequate model. The results show the system can determine liquid volumes with an accuracy of 99.4% using a k-nearest neighbors (k-NN) classification with one neighbor and Manhattan's distance. Moreover, Gaussian process regression using rational quadratic metrics presented a root mean squared error (RMSE) of 0.211 mL.

Keywords: liquid volume measurement; fiber-optic hydrophone; Michelson's interferometer; ultrasound acoustics; active stabilization; machine learning

1. Introduction

Optical fibers have been used worldwide over long distances for ultra-high bitrates in telecommunication systems [1] and for sensor applications [2], including sensing acoustic waves through light modulation [3]. The use of optical fibers as acoustic sensors was first demonstrated in 1977 [4]. Optical fiber sensors (OFSs) have been studied and applied in the industries of petroleum exploration, fuel storage and transportation, industrial manufacturing, military, chemical processing, medical, aerospace, food and civil engineering [5–7].

The oil and gas industry requires control techniques for their vessels, with the aim of increasing production rates, avoiding environmental contamination, mitigating human labor risks, and reducing the costs of plant setup investments (CAPEX), operations, and maintenance (OPEX) [8]. A problem in these industries is the need for continuous monitoring of liquid volumes inside separation vessels working under the presence of dynamic emulsion and foam layers, accumulation of unwanted solids, sand in the bottom and sludge on the walls, turbulence caused by fluid injections, corrosive substances, and explosive atmospheres [9,10].

Ultrasound-based sensors employ acoustic transmitters and receivers immersed in liquids to calculate the acoustic impedance between adjacent layers of fluids [11]. For



Citation: Duque, W.S.; Rodríguez Díaz, C.A.; Leal-Junior, A.G.; Frizera, A. Fiber-Optic Hydrophone Based on Michelson's Interferometer with Active Stabilization for Liquid Volume Measurement. *Sensors* 2022, 22, 4404. https://doi.org/10.3390/ s22124404

Academic Editor: José Manuel Baptista

Received: 21 April 2022 Accepted: 9 June 2022 Published: 10 June 2022

Publisher's Note: MDPI stays neutral with regard to jurisdictional claims in published maps and institutional affiliations.



Copyright: © 2022 by the authors. Licensee MDPI, Basel, Switzerland. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license (https:// creativecommons.org/licenses/by/ 4.0/). reception, piezoelectric hydrophones can be constructed with polymers, such as polyvinyledine fluoride (PVDF), that enable them to provide an output voltage that varies according to acoustic pressure [12]. Besides PVDF, there are ultrasound sensors made with optical fibers [13–22].

Compared to other sensing technologies, optical fiber sensors (OFSs) offer advantages as they are electrically (galvanically) isolated, immune to electromagnetic interference (EMI), intrinsically safe, resistant to chemical corrosion, useable at elevated temperatures, have a wide bandwidth, are capable of multiplexing, are minimally invasive, have good accuracy and resolution, have reduced sizes and weights, their interrogator systems can be installed far from the remotely monitored points, and they do not require any electrical power at the measuring points [8,23,24]. OFSs are precise in measuring pressure, temperature, acoustic fields, strain, torsion, deformation, curvature, force, vibration, acceleration, rotation, humidity, viscosity, and chemical parameters [7,25,26].

In 1977, the authors of [4] first demonstrated the possibility of a direct acousto–optic interaction between an ultrasound field and an optical fiber coil working as an acoustic sensor. The acousto–optic effect is based on the physics principle that a propagating acoustic wave modifies the refractive index of a fluid through pressure variations and densities [20] as a function of time and space. The authors of [4] used 4 m of an SMF optical fiber section length to build a coil of 10 turns and 3.3 cm diameter and submerged it into water so that its optical beam was phase modulated by ultrasound waves ranging from 40 to 400 kHz. The optical fiber coil sensitivity was characterized by light phase shifts whose intensity depended on the acoustic pressure incident on the optical fiber length, with which an MI has been employed [27–29].

A fiber-optic hydrophone (FOH) is an acoustic sensor that uses optical fibers as the sensing element, and can be applied in fields such as oil and gas exploration, earthquake inspection, and underwater object detection [15,30]. FOH is typically implemented based on optical interferometry, which offers greater sensitivity than piezoelectric hydrophones, and they are classed as phase-modulated sensors [7,31,32]. An FOH system measures the light phase changes induced by a particular measurand [7], in which an ultrasound piezoelectric source (UT) works as the transmitter, and an optical fiber sensor works as the receiver [11,33].

Interferometric schemes can be mounted by using commercially available optical components [34]. Four interferometer configurations are usually cited in the literature: Fabry–Perot, Mach–Zehnder, Michelson's, and Sagnac, as they provide good sensitivity, accuracy, large dynamic ranges, and may cover long distances of monitored points [32]. The latter three types are also known as two-beam interferometers [35].

Publications about FOHs have demonstrated their working principles, showing them as an alternative technology to replace piezoelectric hydrophones [36]. FOHs do not need recalibration [37] and present higher bandwidths than conventional piezoelectric hydrophones [38]. Interferometric techniques can be used to improve their sensitivity [39], resolution, bandwidth, dynamic range, and signal-to-noise ratio [40]. A twisted pair made with single-mode fibers has been recently published as a high-sensitivity, broadband ultrasound sensor able to work from 20 kHz to 94.4 MHz [41]. These publications are focused on the technological aspects of FOH.

Other papers present applications and proof of concepts using FOH sensors. As examples, the largest array of FOHs in the world (16,000 sensor elements) is used undersea to permanently monitor oil reservoirs in the North Sea [15], an FOH is used to precisely detect cavitation bubbles under high ultrasound emissions [42], two 32-element line arrays were deployed off the San Diego's coast for object recognition underseas [43], an FOH was used to measure the power of hyperthermia transducers without being damaged [44], and a 50 MHz wideband FOH was developed to measure medical ultrasound fields [45].

To the best of our knowledge, there are no published works describing laboratory prototypes of an FOH system to measure liquid volumes and presenting results of accuracies and RMSE derived from machine-learning algorithms. There are publications

describing the behavior of underwater optical coils, but they do not present results of using these coils to measure liquid volumes [16–19,46–49]. Ultrasound methods to determine multilayers of oil, emulsion, and water in oil tanks are demonstrated in [10,46,50] and show the advantages of FOH, such as contactless distance measurement, low costs, high precision, simple setup, and imperviousness to dusty and smoky environment. However, there were not results regarding liquid volume measurements.

Considering an FOH system as a new generation [15] and a primary application [40] for underwater acoustic sensors and underwater detection [19], the objective of this work is to present a prototype composed of optical fiber coils in the MI's sensor head, with an active stabilization mechanism driving another optical coil wound around a piezoelectric actuator in the MI's reference arm, and a software application that uses amplitudes and phases of detected acoustic signals to predict liquid volume by machine learning algorithms.

Although the isolated functions of the purposed system have been already studied and applied in known applications, the novelty of this proposal is the bringing together of those functions, such as: underwater acoustic wave detection; optical coils as sensing elements; an optical interferometry scheme; an electronic circuit for homodyne detection and active stabilization of noise; another optical coil wound around a piezoelectric actuator; the use of electric phase differences between the acoustic signals inputted and outputted by the sensor; and the definition, testing and comparison of several data models processed by machine learning algorithms, offering new insights and results with high accuracies and low errors for liquid volume measurement.

Section 1 introduces fiber-optic sensors, hydrophones, and related applications. Section 2 presents a theoretical background of acoustic detection, and Section 3 describes the materials and methods employed in the experimental setup. Then, Section 4 presents the results and discussion. Finally, Section 5 ends the paper with conclusions and future work suggestions.

2. Theoretical Background

The MI has been chosen for this work due to its simple configuration and the need for fewer fiber splices and one beam splitter [32,40]. Besides, it accepts flexible geometries and has lower component costs compared to other schemes [7,51]. These characteristics were taken into consideration to choose the MI as the sensor head for this work.

2.1. Michelson's Interferometer

An MI, whose picture is adapted from [7] (p. 371), is presented in Figure 1, in which a 3 dB beam splitter (BS) divides the power of an input coherent laser [31] between two arms defined as optical paths, which end in mirrors that reflect the optical beams back. The optical paths can be free-space air, tubes with gases, vacuum, or optical fibers. The two beams are then recombined by the BS, and a photodetector outputs the phase changes from the interferometric resulting signal.



Figure 1. An FOH based on a Michelson's interferometer.

Considering an MI fed with power P_{IN} , the light wavelength in the medium λ , a wave number $k = 2\pi/\lambda$, and a lossless beam splitter with a division rate of 0.5, the relation between the output power P_{OUT} and the interferometer's optical path differences $\Delta L = L_S - L_R$ can be expressed by [35]:

$$P_{OUT} = \frac{P_{IN}}{2} \left[1 + \cos\left(\frac{4\pi\Delta L}{\lambda}\right) \right]$$
(1)

The first MI arm is called the "sensor head" or "wet arm"; it has a length L_S and is exposed to measurands, such as sound, pressure, temperature, strain, and mechanical vibrations. The other arm is known as the "reference arm" or "dry arm"; it has a length L_R and should be isolated from environment disturbances [7].

A desired measurand induces changes on the sensor head that dynamically change its physical fiber length L_S , leading to a path difference ΔL among the two arms that induces phase differences $\Delta \varphi$ between their light beams, resulting in phase variations that are electronically expressed by a photodetector [7].

Linear polarization of a light beam in one MI arm can be expressed by [31]:

$$\vec{E}(z,t) = E_0 \cos\left[\omega_0 t - \left(\frac{2\pi}{\lambda}\right)z\right],$$
(2)

where \dot{E} is the electrical field of the optical electromagnetic wave, E_0 is its maximum amplitude, ω_0 is the optical wave frequency in radians per seconds, z is the longitudinal space point, and t is time.

Equation (2) shows amplitudes that might be present on the transverse \vec{x} or on the \vec{y} space coordinates, propagating through the longitudinal \vec{z} coordinate. Considering the fiber's physical length L, the light speed in vacuum c_0 , the vacuum wavelength λ_0 , the fiber refraction index n, the wave group velocity $v = c_0/n$, which leads to $1/\lambda = n/\lambda_0$, the electrical field can be expressed by:

$$\stackrel{\rightarrow}{\mathbf{E}} = \mathbf{E}_0 \cos\left[\omega_0 \mathbf{t} - \left(\frac{2\pi \mathbf{n}}{\lambda_0}\right) \mathbf{L}\right],\tag{3}$$

and Equation (4) shows the fixed phase angle ϕ_0 (in radians) of the light beam. The term "nL" is defined as the "optical path length" [52].

$$\phi_0 = \frac{2\pi nL}{\lambda_0},\tag{4}$$

and changes in L or n lead to phase variations, expressed by:

$$\phi_0 + \Delta \phi = \frac{2\pi}{\lambda_0} [\mathbf{n} \cdot \mathbf{L} + \mathbf{n} \cdot \Delta \mathbf{L} + \Delta \mathbf{n} \cdot \mathbf{L}], \tag{5}$$

where $\Delta \phi$ are the incremental phase changes due to incremental length changes ΔL , which induce refraction index variations Δn due to the optical fiber's photo–elastic effect. Further, frequency variations (jitter) from the laser source may also contribute to phase drifts, and Equation (5) can be expressed as a sum of derivatives of the terms involved in the phase drifts [52]:

$$\frac{\mathrm{d}\Phi}{\Phi} = \frac{\mathrm{d}L}{\mathrm{L}} + \frac{\mathrm{d}n}{\mathrm{n}} + \frac{\mathrm{d}k}{\mathrm{k}} \tag{6}$$

Current optical detection systems estimate the average power per unit area and unit time as a measure of intensity or irradiance I_{OUT}, which is proportional to the squared amplitude of the electrical field [53]:

$$I_{OUT} \approx E_0^2 \tag{7}$$

The summation of both arms' light waves $(\vec{E}_T = \vec{E}_R + \vec{E}_s)$ results in an irradiance perceived by the photodetector [31] that can be expressed by:

$$I_{OUT} = 2E_0^2 [1 + \cos(\Delta \phi)], \qquad (8)$$

where $\Delta \phi = \phi_R - \phi_S$ is the phase difference of both light waves' summation.

In another view, this phase difference can also be represented by $\Delta \varphi = \varphi_e - d\varphi_m$, in which φ_e is the bias phase standing for the external disturbances with low frequency and slow drifts with time, affecting both interferometer arms (sensor and reference). The second term $d\varphi_m$ represents the pure physical measurand variations of interest, whose value represents the interferometer's sensor arm [40].

When the interferometer is forced to keep its bias phase ϕ_e around $(2n + 1)*\pi/2$ radians points, it reaches a condition called quadrature state, in which it supplies a better response to the measurand $d\phi_m$. The nature of sinusoidal waves at the quadrature point limits the two-beam interferometers to work under the largest optical phase displacements of half a wavelength. Moreover, when the bias phase ϕ_e is located around zero or on multiple integers of $\pm \pi$ radians, the resulting interferometric pattern jumps between zero and maximum peak values, representing the worst sensitivity points, as the derivatives of intensity to phase are zero at them [31,35,40,54].

Although $\Delta \phi$ may appear to be a stable measurand, the noise present on ϕ_e can be so elevated that the measurand $d\phi_M$ may become impossible to read [31,40]. More details about the MI working as an optical hydrophone for signal intensity (irradiance) and phase (homodyne or heterodyne) measurand detection can be obtained from other studies: [7,15,18,19,25,31,32,40,52,54–56].

2.2. Michelson's Interferometer for Acoustic Wave Detection

Taking into consideration the equations and terms already presented, an MI whose sensor arm is composed of a single-mode optical fiber coil immersed in liquid, within which propagates an acoustic field of pressure P_a and angular frequency ω_a , the interferometer's resulting light beam is expressed by [4]:

$$\dot{\mathbf{E}} = \mathbf{E}_0 \exp i(\omega_0 \mathbf{t}) + \mathbf{E}_0 \exp i(\omega_0 \mathbf{t} + \mathbf{K}_a \sin(\omega_a \mathbf{t}) + \Delta \phi), \tag{9}$$

where K_a is the modulation index that defines the acoustic influence on the phase drifts, and $\Delta \phi$ are the dynamic optical phase differences between both MI arms. For a uniform acoustic field, K_a is expressed below, and L_c is the physical length of the optical fiber coil interacting with the acoustic wave [4]:

$$K_{a} = \frac{2\pi}{\lambda_{0}} \frac{\partial n}{\partial P_{a}} P_{a} n L_{c}$$
(10)

Bucaro et al. [4] also define S_d as constant that defines the detector's sensitivity, which depends on the photodetector's gain G, quantum efficiency q, electron charge $e = 1.602 \times 10^{-19}$ c, Planck's constant $h = 6.626 \times 10^{-34}$ m²kg/s, and on other terms already presented, as expressed by:

$$S_{d} = \frac{2\pi Gqe}{hc_{0}k}$$
(11)

Finally, [4,57] have shown that small values of modulation index K_a will result in the following irradiance equation as perceived by the MI hydrophone:

$$I_{OUT} = 4E_0^2 S_d \left[1 + \cos(\Delta \phi) - \frac{K_a}{2} \sin(\Delta \phi) \sin(\omega_a t) \right]$$
(12)

2.3. Active Stabilization for Homodyne Detection

The phase instabilities inherent to optical interferometers represent a challenge that requires countermeasures to mitigate [35]. In an ideal scenario, the MI works properly when its arms are equal in length (balanced condition) and the reference arm is isolated from external noise [18,54–56]. Thus, MI measurement noise needs to be controlled by adjustment mechanisms [6–8], such as Pockels cells [58], rubber and springs for anti-vibration [28], optical tables, hermetic boxes, thermal baths [56], piezoelectric actuators [18,55], sensing information encoded in a carrier signal, phase generated carriers, and active homodyne demodulation [40]. Despite the solutions aiming to achieve MI stabilization, they are not always continuously stable over time, and, as reported by [28], sometimes noise isolation is "still an art rather than a science".

3. Materials and Methods

This system works based on physics principles in which acoustic wave behavior depends on liquid density, reflections from internal tank walls and other components, and on changes to the acoustic path characterized by changes in the volume of liquid [46].

3.1. Experimental Setup

The experiment setup in Figure 2 shows an FOH system for liquid volume measurement, based on the MI, with an electronic feedback (EF) circuit loop to compensate for external noise from the sensor's measurands. The system outputs a sinusoidal signal from which two values are extracted: amplitude and phase. The phase is the difference between the acoustic field delivered to the UT and the acoustic field detected by two optical coils S1 and S2 placed at the sensor head. For each differential volume of water added to or extracted from the tank, the length of the water (acoustic path) changes, and the internal water surface dislocates. Thus, a new profile of internal sound reflections and backscattering inside the tank creates an interferometric and stationary pattern, knowing that internal water interface works as a mirror to the acoustic waves. Then, the optical coils capture the ultrasound intensities present at their fixed positions, modulating light beams through their photo–elastic (acousto–optic) effects, creating an output signal from which the amplitudes and phases are extracted.

It is not trivial to mathematically equate the components of this setup to reach an expression that infers the water's liquid volume by using the system's output. This setup involves optical and acoustic phenomena taking place simultaneously by an unbalanced MI composed of two optical coils in the same sensor arm with noise stabilized by an EF circuit driving a piezoelectric actuator in the reference arm. The coils are not multiplexed.

3.2. Hardware Components

The setup worked for seven days in an air-conditioned room and water temperatures varying from 22 °C to 24 °C. The test tank was a 1000 mL glass graduated measuring cylinder with 10 mL grading divisions filled with water one milliliter at a time from 440 mL to 1000 mL (561 liquid volume points) with a syringe graduated at 1 mL. The measurements started at 440 mL due to setup arrangements and positioning of the UT inside and in the bottom of the tank. The irradiating face of the UT was positioned at 300 mL volume due to its mechanical size, and the distance of 140 mL between the UT's face and first coil S1 was due to the near field of Fresnel's zone.

A coherent laser light source TN Laser (Teraxion Narrow Linewidth Laser) was tuned at 1550 nm, and its wavelength drift was negligible as it implements internal hardware for electronic current and temperature stabilization. The TNL laser injected 12.5 dBm of light power in the system. The isolator IS (Thorlabs, Newton, NJ, USA) protected the laser source port against undesired returning power. After passing through circulator C1, polarization controller PC1 and the 3 dB beam splitter BS, the light power was equally injected in both MI arms, which ended in Faraday-rotator mirrors M1 and M2. The MI's reference arm was composed of an optical coil of 11 m [55] made with SMF fiber (Corning SMF-125/9, only with acrylate protection) wound around a piezoelectric actuator PZT (Thorlabs APF705) responsible for stabilization. The sensor arm's coils S1 and S2 were both identical and made of the same fiber type, with an outer diameter of 2.4 cm, and 25 fiber turns, resulting in 188.5 cm of total fiber length per coil. In total, considering all fibers used, component connections, installation arrangements and coils, each MI arm achieved a length of around 15 m. Single mode fibers were used because the behavior and analysis of multimode interferometers are difficult to control [56].



Figure 2. Fiber-optic hydrophone (FOH) system setup. TNL Laser at 1550 nm; IS—optical isolator; C1 and C2—optical circulators; BS—beam splitter; PC1 and PC2—polarization controllers; PD1, PD2 and PD3—photodetectors; M1 and M2—Faraday's mirrors; PZT—piezoelectric actuator wound by an optical fiber coil; S1 and S2—optical fiber coil sensors; UT—ultrasound transducer; EF—electronic feedback loop circuit; OP—oscilloscope; SG—signal generator; A—3 dB amplifier; Application is software developed in MATLAB; liquid volume measurement tank.

All optical components, such as the isolator, circulators, beam splitter, Faraday mirrors and PZT with its optical coil, were installed inside boxes (non-hermetic) with the goal of mitigating external influences, such as temperature variations, random pressure fluctuations, sounds, air flows, or mechanical stress on fibers and connectors. The optical components and photodetectors were mounted over an optical bread board with a rubber foot to mitigate mechanical vibration and vibration on the table where the components were situated. These disturbances were controlled as they could decrease measurand accuracy [18,55,56].

The polarization controller PC1 was placed at the MI input to adjust the polarization and improve the contrast [56]. Following the same idea, PC2 was installed at the MI output, which delivers the sensor's measurand signal to the photodetector PD3. The two circulators C1 and C2 provided the required optical paths [55] to photodetectors PD1, PD2, and PD3 (Thorlabs PDA30B-EC). This optical arrangement guaranteed that the optical signals input to PD1 and PD2 were 180° out of phase. While both photodetectors PD1 and PD2 were only working for the stabilization mechanism, PD3 was uniquely responsible for supplying the sensor system's output signal. The EF circuit was fed by PD1 and PD2, and its output drove the PZT, which stretched the optical coil of the MI's reference arm, keeping the interferometer locked in quadrature state and providing a final sensing signal with less interference through PD3. On the electronic side, the Signal Generator SG (BK Precision 4053) was set to provide a 100 kHz continuous sinusoidal wave (CW) on both its output channels, CH1 and CH2, with the same phases but different peak-to-peak voltages of 10 Vpp and 1.8 Vpp, respectively. During the preliminary tests, a range of UT frequencies were evaluated from 50 kHz to 1 MHz, and an adequate response range of voltage and phase variations occurred at 100 kHz. The lower UT (Precision Acoustics Unfocussed 1 MHz, 23 mm diameter and 75 mm focal distance) frequency was used to create higher wavelength sounds in the water, supplying a better response to the volume variations of 1 mL.

The SG-CH1 signal was amplified by the Juntek DPA-2698 (BW 10 MHz and 3 dB gain) amplifier (A), which output a 20 Vpp sinusoidal CW signal of 5 W to the UT. The second port of signal generator SG-CH2 was connected to the CH1 input of the Oscilloscope OP (Tektronix Open Choice MDO 3012) to work as a reference for FOH system phase recovery. The UT was positioned in such a way that the ultrasound waves were irradiated upwards through the liquid, and their wavefronts were parallel to the plane of the optical coils S1 and S2. The FOH output was injected into the second oscilloscope port (OP-CH2).

Both OP ports (CH1 and CH2) were configured to supply AC coupling and BW of 20 MHz, making them more adequate for the involved frequencies and to cut DC power. The OP's Fourier-transform function allowed seeing one channel in both time and frequency domains simultaneously, and the signal amplitude of 1.8 Vpp from SG-CH2 was chosen to help with adjustment during setup. Last, the MATLAB application (MA) [59] read the signals from both oscilloscope ports and extracted two measurands: amplitude in volts and phase in degrees. This phase value was calculated by MA and stood for the difference between the original signal injected into the UT (interferometer input) and the resulting signal delivered by PD3 (main MI output). This output was then digitalized by the oscilloscope at a sample rate of 100 MS/s, the samples were read in vectors of 10 thousand points of double type variables, which were passed through a software band-pass filter and then to a Fourier transformation, from which the values of amplitude and phase were registered in a database.

3.3. Fresnel's Acoustic near Field

A Fresnel's near field was kept free between the UT surface and the first coil S1. The acoustic's near field was a Fresnel's space that should not have been obstructed by any object so that the acoustic field [12] emitted by the UT would not be destroyed or severely attenuated, which would have compromised the acoustic detection by the optical coils in the MI's sensor head. The calculation of this field considered the UT's diameter (D = 23 mm), the ultrasound frequency (f = 100 kHz), and the wave speed in water [60]. At lower temperatures, the larger the near field; the worst case of T = 22 °C was used to calculate the water's sound speed s_{wa} [60] and the near field distance F_{nf} [12] as follows:

$$s_{wa} = 1404.3 + 4.7 \times T - 0.04 \times T^2 = 1488.34 \text{ m/s}$$
 (13)

The near field for a cylindrical UT is expressed by:

$$F_{nf} = \frac{\left(\frac{D}{2}\right)^2}{\lambda_{wa}} = \frac{\left(\frac{23 \times 10^{-3}}{2}\right)^2}{\lambda_{wa}} \therefore \lambda_{wa} = \frac{s_{wa}}{f} = \frac{1488.34}{100 \times 10^3} \therefore F_{nf} = 8.9 \text{ mm}$$
(14)

For the test tank of this experiment and $\lambda_{wa} = 14.9 \text{ [mm]}$, the distance $F_{nf} = 8.9 \text{ [mm]}$ is equivalent to the volume range $F_{nf} = 26.3 \text{ mL}$. Considering that the UT surface was placed at the volume of 300 mL, the near field for f = 100 kHz and $T = 22 \degree \text{C}$ is found in the range from 300 mL to 326.3 mL and, therefore no objects, not even an optical coil sensor, should be put in this region.

As higher ultrasound frequencies were also used during the preliminary tests, the free water volume of 140 mL purposely left between the UT surface and the first S1 coil allowed the use of a maximum frequency of 520 kHz without obstructing the near field.

3.4. Electronic Feedback Loop Circuit

In this work, an active and self-compensate mechanism for the MI was adopted to control the environment disturbances and offer better stability to the measurand data perceived by the sensor head. It was based on the work developed by [18,55], who implemented the electronic feedback (EF) loop circuit shown in Figure 3.



Figure 3. Schematic of electronic feedback (EF) loop circuit for Michelson's stabilization.

The EF was composed of four operational amplifier chips (U1 to U4), a dual comparator chip U5, a dual flip-flop chip U6 (CD4013BC), one four-electronic keys chip U7 (CD4016BCN), and other discrete components such as resistors (18 units), capacitors (2 units), and diodes (3 units). This EF circuit has two inputs, whose signals are received from the photodetectors PD1 and PD2, and one signal output, which drives the PZT. The output values of PD1 and PD2 were matched through dB gain adjustment knobs of the photodetector's hardware (Thorlabs PDA30B-EC).

Some minor changes were made to the original author's electronics and optical schemes published by [55] to simplify the setup, but without affecting any original functionality of the circuit. Both input low pass filters were not electronically implemented, as they were provided by the oscilloscope input ports and by a software high-pass filter implemented in MATLAB. Furthermore, the DC power supplies were simplified to use the values of +10 and -10 volts, cutting the need for +5/-5 and +15/-15 VDC inputs. Finally, the four U7 electronic switches were connected in parallel to decrease their internal resistance and to allow proper PZT operation.

The active stabilization's EF loop acts as a homodyne demodulator that nullifies as much as possible the output of the differential amplifier U3, locking the MI operation around the vicinities of the quadrature point [40]. The interferometric signals output by PD1 and PD2 (electronic currents) were 180° out of phase and input to the EF circuit for differential detection.

$$I_{PD1} = I_0 [1 + K_0 \cos(\phi_s + \phi_d)],$$
(15)

$$I_{PD2} = I_0 [1 + K_0 \cos(\phi_s + \phi_d + \pi)],$$
(16)

where I_0 relates to the system input power, K_0 is related to the interferometric fringe visibility, ϕ_s is the static differential phase, and ϕ_d is the disturbances' differential phase.

After PD1 output is inverted by U1, and, together with PD2 output, both signals are differentially combined by the converter U2, the following signal is seen on the U2 output, where K_2 if the conversion gain of U2:

$$U2_{OUT} = K_2 \times \cos(\phi_s + \phi_d) \tag{17}$$

After $U2_{OUT}$ is integrated by U3, the new output signal is presented below, where K_3 is the conversion gain of U3:

$$U3_{OUT} = K_3 \times \sin(\phi_s + \phi_d - \pi/2) \tag{18}$$

As the perfect quadrature point is defined by $\phi_s + \phi_d = \pi/2$, and the interferometer is forced to be kept around the quadrature state, then U3_{OUT} will tend to zero, knowing that, the differential phase will be:

$$\Delta \phi = \phi_{\rm s} + \phi_{\rm d} - \pi/2 \therefore \Delta \phi \cong 0 \tag{19}$$

However, this absolute zero is not supported, and in the vicinity of the quadrature point, differential phase $\Delta \phi$ drifts assume small values, allowing the approximation below, considering that $\sin(\Delta \phi) \cong \Delta \phi$ for small values of the angle $\Delta \phi$:

$$U3_{OUT} = K_3 \times \sin(\Delta \phi) \therefore U3_{OUT} \cong K_3 \times \Delta \phi, \tag{20}$$

This simplification represents a practical assumption when a sine or cosine math function falls inside a cosine or sine formula, such as the term "exp $i(\omega_0 t + K_a \sin(\omega_a t) + \phi_0)$ " in Equation (9). This assumption is considered valid for $\Delta \phi \ll 1$ and $\Delta \phi \leq 0.1$ radians as described as acceptable by [52]. In fact, it avoids the need to consider many terms of Bessel's functions of the first kind, as the first two terms of their coefficient family, $J_0(\Delta \phi)$ and $J_1(\Delta \phi)$, represent 98% of the homodyne power spectrum energy [52].

Therefore, U3_{OUT} is quasi-linear at the quadrature state [18]. It is used as the correction signal that drives the PZT, making displacements and strain changes in the optical fiber coil wound around it, keeping the interferometer working close to the quadrature state.

The EF circuit also tracks the external noise through an equivalent voltage in capacitor C1. The circuits U5.1 and U5.2 compare C1 with the reference voltage defined by the potentiometer POT (R18), and, when they are equal, the EF switches the C1 voltage to zero by an up-pulse to flip-flop U6 that keeps the four U7 switches closed and discharging by the time $\tau = R19 \times C2 = 0.47$ ms. The reference value of +5 volts is selected with R18 (potentiometer) so that the limits of +5/-5 volts of accumulated phase drift errors are enough to protect the PZT and other electronic components, respecting their technical datasheets [18]. In general, the EF circuit can be regarded as a high-pass filter with a cutoff angular frequency f_{EF} in hertz, as expressed by [55]:

$$f_{EF} = \frac{K_p K_d}{2\pi \times R8 \times C1}$$
(21)

The voltage-phase coefficient K_p rad/V for the PZT actuator and its optical fiber coil and the phase-voltage coefficient K_d V/rad for photodetectors have been experimentally measured by driving the PZT with a low-frequency triangular wave [55], allowing calculation of f_{EF} Hz, and are further presented in the results section.

3.5. Sensitivity, Resolution, Bandwidth and Other Technical Characteristics

The purposed setup is composed of different elements, both optical and electric, each with specific technical parameters for power, frequency, bandwidth, sensitivity, resolution, internal noise sources, and others [40]. The proposed FOH system's sensitivity, resolution, and bandwidth are dependent on the setup's arrangement and on its components' intrinsic characteristics, whose values are examined in the results section, following collected data series and their derivatives of phases and amplitudes.

Although it has not been the scope of this work to measure isolated parameters of the optical coils built for the MI's sensor head, other studies have already demonstrated a sensitivity of $-116 \text{ dB } re \text{ rad}/\mu\text{Pa}$ in the bandwidth of 10 Hz to 2 kHz, phase noise of $-102 \text{ dB } re \text{ rad}/\sqrt{\text{Hz}}$, and pressure noise of 14 dB $re \mu\text{Pa}/\sqrt{\text{Hz}}$ [15]. Another study shows a sensitivity of $-170 \text{ dB } re \text{ rad}/\mu\text{Pa}$ and noise floor of 50 dB $re \mu\text{Pa}/\sqrt{\text{Hz}}$ below 1 kHz [40]. These parameters vary with the type and length of fibers exposed to the acoustic waves, coil sizes and their layout, number of fiber layers, acoustic wave frequency and direction, mechanical and elasto–optic properties of fibers, and other parameters [51,61–63].

3.6. Data Collection and Database Structure

The MA developed for this experiment was connected online to all setup equipment and took twenty measurements of amplitude and phase per each milliliter of liquid volume, for a total of 11,220 readings (20×561) that were stored in a database. Values of amplitudes (volts) were rounded to 2 digits, and phases (degrees) were rounded to zero digits to the right of the decimal point to allow the system to calculate the modes (most frequent values) of both measurands.

Each reading was registered as a tuple of data object (lines) having 64 attribute fields (columns) of information. Among these fields, the tuple had point values of amplitude and phase, along with modes, means, standard deviations, and outlier indicators. A tuple also carried fields for both intermediate and final values of modes and means. Twenty measures composed a group, and the intermediate values were calculated based on the measurements taken until the moment of a point collection. For example, the 12th measurement of a group registered intermediate values of mean and mode for the twelve measures taken until that moment. The last measure of a group was calculated after removing the interquartile outliers. In this work, these outliers were values greater than 1.5 times the interquartile size, above the upper quartile (75%) and below the lower quartile (25%), for a measurement group. The interquartile method was applicable for data both normally distributed or not. The database also registered a field to show outliers that were less than 10 or above 90 percentile ranges. These two strategies are referred to in this text as "interquartile" and "percentile" outliers, respectively.

As temperature changes acoustic properties of liquids, each object tuple also had a field that registered water temperature, which was provided by an electronic sensor installed inside the test tank and connected to the MA. This field was used by the machine learning algorithms with the action of merely adding it or not among the columns of attributes selected to compose a data model to be assessed.

Further, no phase or amplitude values from repetitive cycles of increasing and decreasing volumes were registered in the database, despite the fact that this repeatability was seen during the setup tests.

3.7. Phase and Amplitude Data Series Analyses

This work presents data series analyses of amplitudes and phases, including their derivatives in relation to liquid volumes. Standard deviations and small linear intervals of these series were evaluated. All graphs show results before and after the S2 coil was submerged. The data series include the mode values of each measurand. First, the phase and amplitude were individually analyzed, and further both were put together in a two-dimensional Cartesian space. A specific phase analysis included three adjustment actions, after which an unwrapped phase characterization was obtained for the FOH system.

A limitation in the arctangent function executed by homodyne detectors, where sine and cosine signals are converted to phase, is that the phase can change a maximum of π radians between samples. So changes greater than π represent ambiguity as to whether the signal travelled a full clockwise or counterclockwise lap around the four quadrants [64]. Therefore, an absolute change higher than 180° between two adjacent milliliters may be a discontinuity caused by a new wrap cycle in the domain of 360°, influenced not only by the intrinsic nature of the system but also by noise that was still present in the measured data despite the stabilization mechanism. Under this condition, predictive algorithms were used to improve the π limitation [40] as performed by other publications about phase measurands from optical interferometers [30,64,65].

3.8. Three Actions of Phase Series Adjustments

Taking into consideration the aspects of phase detection reported in the earlier topic, three actions were applied on the phase data collected in this work.

Action (a)—Fixing Phase Discontinuities with 360° Rotations

In action (a), all absolute variations above 180° between two adjacent milliliters were rotated by 360° . For example, if a liquid volume point is -270° and the earlier point was $+45^{\circ}$, then a 315° absolute variation is detected and, therefore, the -270° point is changed to $+90^{\circ}$ after a 360° rotation. If this new value also caused new further absolute variations above 180° , the respective liquid volumes were rotated as well.

Action (b)—Unwrapping the Phase Series

In action (b), the phase data series was unwrapped, in which the angles were rotated by $\pm 180^{\circ}$ based on the following thresholds vector: $(-180^{\circ} - 90^{\circ} - 60^{\circ} - 30^{\circ})$. The minus signal of a threshold value works only on up–down variations, and the plus signal on down– up variations. The algorithm ran on each phase difference between two adjacent milliliters, and if it was higher than the threshold value and was in the signal direction of the threshold signal, the point and the remaining series were rotated by 180° . Up–down movements were rotated by $+180^{\circ}$ and down–up by -180° . In the end, these rotations performed an unwrap operation in the whole phase series, respecting the arctangent function properties as the earlier and the updated angles still kept the same tangent values. Different values of this vector were assessed, and it was found that positive thresholds (high down–up movements) did not affect the unwrapping process.

Action (c)—Excluding Phase Points with Deviations Above 60°

In action (c), as a strategy for phase noise treatment, a phase point was excluded when the standard deviation in the measurements group of liquid volumes was greater than 60° during data collection.

3.9. Fitting Model for Phase Characterization

After performing the three actions of phase adjustments, the best result was selected based on the greater R-squared (R^2) and minor root mean square error (RMSE) according to the fitting model assessed for the final series of the unwrapped phase. Among the models evaluated, there were polynomials, exponentials, Fourier series, Gaussians, power series, and sums of sines.

RMSE and RMSE% are calculated by Equations (22) and (23), respectively, where Y_{Ti} and Y_{Pi} are the true and predicted values, respectively, of the ith term in the phase series composed by N terms.

$$RMSE = \sqrt{\frac{1}{N} \sum_{i=1}^{N} (Y_{Ti} - Y_{Pi})^{2}},$$
(22)

RMSE% =
$$\sqrt{\frac{1}{N} \sum_{i=1}^{N} \left(\frac{Y_{Ti} - Y_{Pi}}{Y_{Ti}}\right)^2 \times 100\%}$$
, (23)

3.10. Machine Learning Algorithms

This work employed machine-learning algorithms under supervised learning to predict liquid volumes from amplitudes and phases of acoustic waves. Two distance-based techniques were tested, Rocchio and k-NN [66], with five distance metrics: Euclidean, Mahalanobis, cosine distance, cosine similarity, and Manhattan. Rocchio is a classifier based on the nearest centroid defined by the training data, and k-NN is based on the k-nearest neighbors to the training data.

In total, six scenarios were evaluated, and each scenario was defined by a specific choice of object tuples (lines) and attributes (columns) from the $11,220 \times 64$ matrix of collected data. A group of selected tuples defines a dataset, and the group of selected attributes defines a data model. In this case, a data model can include, for example, water temperature, so as to see the influence of this field in the results. Further, each scenario has its own criteria to separate the training from the testing datasets. As an example, a training dataset can be composed only of the twentieth object tuple of the liquid volume points

(mode centroids), and the testing dataset by the remaining objects, or the training set can be composed of a random selection of 75% of all tuples, and the testing dataset includes the remaining 25%. In addition, other characteristics, such as distance metrics used and the "k" number of neighbors in k-NN were assessed.

When the training dataset is composed of one tuple standing for the centroid of a milliliter, both Rocchio and 1-NN reach the same accuracy. So, in these cases, the 3-NN is presented, while other cases used 1-NN as it delivered better accuracy. For all scenarios evaluated, both training and testing datasets were normalized to avoid distortions caused by the scales of the attributes. Each liquid volume point was defined as a class (label) with which a new aleatory sample of phase and amplitude was processed to obtain the respective liquid volume. The data were balanced, as the classes have the same number of samples, although they can vary a little with scenarios that excluded outliers.

3.11. Gaussian Process Regression

In addition to the use of classification algorithms, this work also assessed machinelearning regression versions based on the same fitting models used to characterize unwrapped phase data series. However, instead of testing only the mode values of amplitudes and phases, the regression models with machine-learning were applied to all database points with the same random selection of 75%/25% to define the training and the testing datasets, respectively. The best result in terms of R2 and RMSE was obtained from Gaussian process regression (GPR), whose values are presented in the result section of this paper. GPR assumes that the target variables are Gaussian distributions and makes use of covariance matrices derived from data attributes to predict a fitting model [66].

4. Results and Discussion

4.1. Sensitivity, Resolution, Bandwidth, and Other Technical Characteristics

Regarding acoustic bandwidth, the system was designed to operate at the fixed frequency of 100 kHz with the ultrasound transducer able to work at the maximum frequency of 1.5 MHz, and Figure 4 shows an experimental result of a 6 dB bandwidth at 700 kHz, ranging from around 30 kHz to 730 kHz, inside which specific frequencies of 80 kHz, 220 kHz, and 340 kHz presented high peak values in the response curve. This response was computed under a fixed liquid volume, as different volumes present different voltage amplitudes for a fixed frequency. The choice of the FOH system's fixed operation frequency took into consideration experimental bandwidth, the Fresnel's zone restrictions, and the longest possible acoustic wavelength in the water.



Figure 4. Experimental FOH system's ultrasound frequency response curve.

The EF loop circuit cutoff bandwidth was calculated following [55], and Figure 5 shows the oscilloscope graphical results used to obtain the sensitivity $K_p = \frac{31.5\pi}{4.24} = 23.34 \text{ rad/V}$ and $K_d = \frac{1.06}{\pi/2} = 0.67 \text{ V/rad}$ after driving the PZT with a 50 Hz and $V_{pp} = 4.24 \text{ V}$ triangular wave. The only requirement to choose the triangular wave's frequency was to make it as small as possible in order to allow counting the number of cycles in the resulting sinus modulated wave. Therefore, the experimental high cutoff frequency of the EF circuit loop resulted in $f_{EF} = \frac{23.34 \times 0.67}{2\pi \times 100 \times 10^3 \times 4.7 \times 10^{-6}} = 5.33 \text{ Hz}$, which was enough to mitigate most environmental noise. A replacement of capacitor C1 from 4.7 µF to 1 µF raised this value to 25.07 Hz, which is close to the value experimentally found by [55].



Figure 5. Coefficients K_p (PZT with optical coil) and K_d (photodiodes) experimentally obtained.

4.2. Original Amplitude and Phase Data Series Collected

The original collected values of amplitudes and phases per each liquid volume point are plotted in Figure 6. As a homodyne detector (phasemeter), each amplitude in volts stands for the optical phase displacement between the signals from the MI's arms, related to a specific liquid volume, and each phase point is the electronic acoustic phase difference between the input and the output signals from the FOH system. Phasemeters recover the phases from the four-quadrant arctangent according to the ratio between two quadrature signals (a sine and a cosine). The relationship between the expected phase and the phase effectively measured should be linear, but there are often distortions caused by spurious effects in the optics or in the signal processing that result in the nonlinearities and periodic errors [35] seen in Figure 6.

Specific characteristics can be noted from the originally collected data. Both amplitude and phase series are nonlinear with liquid volumes; they are not intrinsically correlated to each other; there are different peaks and valleys when only S1 is under water than when both S1 and S2 are submerged; below S2, the phase varies around the limits of -45° to $+45^{\circ}$, covering a range of 90° as limited by the quadrature; above S2, the phase measurand shows maximum and minimum values that touch the limits of -270° and $+90^{\circ}$, covering a range of 360°; above S2, the amplitude window of peaks and valleys dislocates down; both series present limited linearity intervals with liquid volumes, as seen in Figure 7; the amplitude shows quadratic behavior at volumes of 820, 890, 920, 940, and 970 mL; the phase measures show discontinuity jumps near the volumes of 650, 690 to 710, 740, 860, 920, 940, and 990 mL; and the first measured phase at 440 mL presented a value smaller than -180° .



Figure 6. Original amplitude and phase mode values collected per milliliter.



Figure 7. Phase measurand with correction of discontinuity jumps $\geq 180^{\circ}$.

These results confirm nonlinearities, differences in using one or two optical coils as sensing elements, attenuation of acoustic fields as the volume of water increases, and that phase differences between the optical signals in the MI arms are not related to the phase differences between the acoustic signal input and output from the system.

4.3. Phase Measurand Analysis

Three actions were applied to the collected phase data in order to: (a) fix the discontinuities; (b) unwrap the data series; and (c) mitigate the noise.

In action (a), after all rotations, the new phase profile is shown in Figure 7, from which it is noted the effect of dislocating the whole data series from the limits of $-270^{\circ}/+90^{\circ}$ to a new range of $-180^{\circ}/+180^{\circ}$.

For comparison with the original data presented before, Figure 8 shows the original amplitudes and the new phase series corrected with the 360° rotations.



Figure 8. Original amplitudes and corrected phases.

A derivative analysis of both amplitude and phase measurands is presented in Figure 9, with common phase sensitivities reaching 20° degrees/mL (349 milliradian/mL) and amplitudes in 70 mV/mL. The dashed lines are one standard deviation over means for both measurands, separated by liquid volumes below and above the S2 coil. Comparing variations due to the influence of the S2 coil, the amplitude deviation decreased from 44 mV to 31 mV, while the phase increased from 24° to 37° .



Figure 9. Derivatives of amplitudes and corrected phases.

That is, the presence of S2 decreased the sensitivity of amplitude (homodyne detection related to the optical phase differences between the signals inside the MI arms) but increased the sensitivity of the acoustic phases (related to the electrical phase differences between the signal input to the UT and the measurand signal delivered by PD3). Based on that result, future analysis would take S2 out of the system and verify if coil S1 alone could detect different series of amplitudes and phases, including their derivatives.

Figure 10 shows plain linear intervals, with numbers at the ends of straight lines being the first and final liquid volume of each interval. This figure also shows that the majority of phase points have their mode values ("x") close to their mean values ("+"), which is a



characteristic of normal distributions, although neither normal fitting test was performed over the data, as it was not a goal of this work.

Figure 10. Liquid volume intervals with linearities with phase.

Figure 9 shows that after executing action (a), the phase derivative series are mostly confined in the limits of $-20^{\circ}/20^{\circ}$ with one standard deviation of 37° . So, a maximum admissible standard deviation error of 60° for a point in action (c) is a conservative value, although other values have been evaluated. So, action (c) caused the elimination of 4% of collected data, and Figure 11 shows the resulting series after each one of the three phase actions were executed. The bottom of Figure 11 also shows room and water temperatures, both measured during data collection days represented by small circles on the graph.



Figure 11. Three actions of phase series adjustments.

According to Figure 12, while the original phase derivatives reached a six-standard deviation of 290°, the final adjusted phase series ended with a six-sigma deviation of 82°. This result was achieved after the execution of three sequenced steps of phase adjustments, whose algorithms improved upon the π constraints of angle measurands [40]. In fact,



those steps detected and cleaned phase jumpers between adjacent milliliters with absolute values higher than 180° and unwrapped the phase series after removing phase points with deviations equal to or higher than 60° from the original data.

Figure 12. Derivative analyses of phase signals before and after corrections.

Derivative analysis showed the improvements achieved by these proposed phase adjustment algorithms, which had the goal of finding a final phase characterization for the FOH system. Those actions were based on parameters that could be previously configured, allowing the MA to apply them as long as new data were collected online.

Finally, Figure 13 shows the phase characterization after fitting the unwrapped phase with a sum-of-sines of nine terms, resulting in R2 of 0.9994 and RMSE% of 4.15%. The model was configured to run with least absolute residual method and nonlinear least squares.



Figure 13. Hydrophone system's unwrapped phase characterization.

The sum-of-sines fitting function is expressed by:

$$L_{ml} = \sum_{i=1}^{9} a_i * \sin(b_i * \Delta \varnothing_s + c_i), \qquad (24)$$

where L_{ml} is the predicted liquid volume in milliliters, and $\Delta \varnothing_s$ is the angle detected and processed by the FOH system's application after the three phase treatments are applied. The coefficients a_i, b_i, and c_i of the adjusted function are listed in Table 1.

i	a _i	b _i	c _i
1	2419	0.7729	0.7455
2	1087	1.66	-2.039
3	288.9	3.401	0.8193
4	102.6	5.768	-0.8964
5	29.65	9.122	-2.307
6	5518	18.81	-1.821
7	5517	18.81	1.317
8	-4.334	13.82	9.008
9	13.23	45.2	-0.7065

Table 1. Coefficients of sum-of-sines fitting function.

4.4. Amplitude Measurand Analysis

The amplitude and its derivative series have already been presented in the earlier section to allow their comparison with the phase series. So, a next analysis is to show the presence of linear intervals, as shown in Figure 14. This figure also shows that most amplitude points have mode values quasi-equal to mean values, what is also a characteristic of normal distributions.



Figure 14. Liquid volumes with linear intervals on the amplitude series.

Figure 15 shows the derivatives of amplitude to liquid volumes, with limited ranges of linear behavior.



Figure 15. Liquid volumes with linear intervals on the derivative of amplitude series.

The amplitude analysis shows that the collected data are nonlinear, and it is not obvious how to derive a math function to fit these data. In this sense, the next section presents an analysis using both phase and amplitude mode points.

4.5. Phase and Amplitude Analysis

This topic evaluates both amplitude and phase as coordinates of a 2D Cartesian space. Figure 16 uses the corrected phases previously presented in Figure 7, and Figure 17 uses the unwrapped phases presented in Figure 13. Both graphs have the phase measurand in the *x*-axis and the amplitude measurand in the *y*-axis, with point coordinates defined by their mode values. The integer numbers plotted over the sequenced points are the respective liquid volumes in milliliters. Considering the substantial number of measured points, not all values were plotted to preserve an adequate graphical visualization by the reader.



Figure 16. The 2D grade of amplitudes and corrected phases (only action "a" applied).



Figure 17. The 2D grade of amplitudes and unwrapped phases (after action "c" is applied).

Figure 16 shows points standing for a linear sequence of liquid volumes, but with the evolute as curves of amplitude and phase coordinates making clockwise turns as long as the liquid volume increases linearly, forming patterns like "snail paths" that are overlapped by other paths, showing nonlinear behavior of the FOH system. This pattern also shows spaces with a high concentration of liquid volume points and empty spaces in the vicinity. There is also a difference in space point occupation when S1 is submerged (liquid volumes below S2) compared to when both S1 and S2 are underwater. In fact, the unwrapped phases are seen only with measurements above S2.

4.6. Rocchio and k-NN with Different Distance Metrics

This topic presents six scenarios of data analysis. Scenarios one to four used the phase adjusted to the range of $-180^{\circ}/+180^{\circ}$. Scenario five used the unwrapped phase, and scenario six used the original phase without any adjustments.

Figure 18 shows the results of first scenario that evaluated the mode values of phase and amplitude related to a liquid volume point. The mode values were used as the training dataset, and they were evaluated against themselves as testing data. In this scenario, an accuracy of 100% was expected, but the accuracy of 96.8% shows that 3.2% of the centroids overlap; that is, although they stand for different liquid volumes, they have the same coordinates of amplitude and phase. This result corroborates the behavior of Figure 16 in which the "snail paths" overlap each other more than one time. This also shows that Rocchio is not applicable if data are composed of centroids of mode values. Additionally, k-NN can be seen as useless with an accuracy of 31.2%. The small circles in the graphs supply a visual notion of the wrongly classified points.



Figure 18. Rocchio and k-NN accuracies with centroids of mode values.

This first Rocchio result shows a trade-off in which the need to round numbers to allow the use of mode values has the collateral effect of increasing the probability of centroids overlapping. So, the number of digits to the right of the decimal points is an input parameter to the system. The fewer digits, the higher probability of overlapping. Based on the results of this first scenario, new data models and new selections of training and testing data were defined to improve analysis.

A second scenario used the same data model as the first one, but with modes replaced by means, and Figure 19 shows an accuracy of 100% for Rocchio and 29.8% for 3-NN.



Figure 19. Rocchio and k-NN accuracies with centroids of mean values.

If both modes and means are used in the data model as a third scenario, the Rocchio and 3-NN accuracies are 100% and 31.4%, respectively. A partial conclusion here is that both training and testing datasets, besides the data models, still need to be improved. Figure 20 shows a fourth scenario in which the training data are the centroids of modes and means of phase and amplitude, and the testing data are all the remaining points collected by the experimental setup excluding the centroids and the interquartile outliers.

The fourth scenario is more data intensive than the third, as it used more tuples as testing data, resulting in an accuracy of 57.8% for Rocchio and 25.0% for 3-NN. Additionally, 5-NN's accuracy is 15.8%. In this sense, it is possible to note that training datasets composed only of centroids are poor data models, as their low distances and overlapping points are not adequate for distance-based algorithms.

Therefore, a fifth scenario was defined, in which the data model was improved, and its attributes were composed of point values, means, and modes (intermediate and finals) for both amplitude and phase, complemented by the water temperature. The fifth scenario considered the unwrapped and characterized phase data presented in Figure 13.



Figure 20. Rocchio and k-NN accuracies with complete datasets excluding outliers.

The choice of object tuples (lines) was aleatory to define the training and the testing datasets with the respective proportion of 75%/25%. Initially, no outliers were excluded. The k-NN was defined to one nearest neighborhood (1-NN) as it presented better results than when using three or more neighbors for all distance metrics.

As a result, for Euclidean distance, 1-NN reached an accuracy of 91.6%, and other k-NNs with three or more neighbors presented lower accuracies (3-NN = 87.9%; 5-NN = 85.3%; 9-NN = 80.2%; 15-NN = 71.6%; and 20-NN = 64.0%), reinforcing that 1-NN is always better.

Figure 21 shows the resulting accuracies for different distance metrics used with Rocchio and k-NN, in which 1-NN with Manhattan distance reached an accuracy of 97.0%. The cosine similarity metrics resulted in accuracy of 0% (zero) for both algorithms and for all scenarios evaluated so far.



Figure 21. Cont.



Figure 21. Fifth scenario. This figure has ten graphs distributed in two columns and five lines. Graphs on the left column are Rocchio and on the right are 1-NN. Each line stands for a different distance metric. Each graph title has the name of the distance used, the resulting accuracy, and number of tested samples.

These distance-based analyses could have stopped with the fifth scenario, as an accuracy above 95% was found. However, perceiving the result of Manhattan's distance metrics in the fifth scenario, a sixth and last scenario was proposed with the goal of testing new attribute models and data sources only using that metric. The data sources were differentiated according to the three phase adjustment actions previously presented. A phase series could be (i) original; (ii) adjusted to $-180^{\circ}/+180^{\circ}$; (iii) unwrapped; or (iv) unwrapped with exclusions of deviations above 60°.

Testing revealed the best accuracy found for the sixth scenario occurred when the data model attributes and data tuple selection were performed the same as for the fifth scenario, but with the original phases as collected from the experiment without any adjustments. Fine-tuning of the data model was also applied, in which different combinations of interquartiles and percentiles were evaluated using the fields of outlier indicators originally registered

by the FOH's MA application. As there are four variables, sixteen combinations of two outliers and two measurands were evaluated, and their results are presented in Table 2. When an outlier exclusion is applied, it is shown by an "x" in the table. Figure 22 shows the best graphic results derived from theses analyses.

	Amp	litude	Ph	ase		Manh	attan
Test	Quartile	Percentile	Quartile	Percentile	Samples	Rocchio	1-NN
1	х	х	x	x	1799	96.1%	99.1%
2	х	х	x		2170	97.3%	99.4%
3	х	х		х	1801	96.3%	98.7%
4	х	х			2226	96.0%	99.3%
5	х		Х	х	2120	96.9%	99.1%
6	х		Х		2551	96.7%	99.0%
7	х			х	2122	96.3%	99.2%
8	х				2622	95.9%	99.2%
9		х	Х	х	1811	96.3%	99.0%
10		х	Х		2186	96.1%	98.9%
11		х		х	1814	95.9%	99.0%
12		х			2244	96.1%	98.8%
13			Х	х	2240	96.1%	98.8%
14			Х		2697	95.9%	98.5%
15				х	2244	95.8%	98.6%
16					2805	92.3%	98.0%

Table 2. Sixth scenario results for Rocchio and 1-NN with Manhattan and outlier selections.



Figure 22. Rocchio and 1-NN accuracies with Manhattan distances over original data collected.

Although Test 2 of Table 2 presented the greatest accuracy of 99.4% among the 16 tests, Figure 22 shows that Test 1 also has a high accuracy of 99.1% but a cleaner graph, in which the distances between the correctly and the wrongly predicted liquid volumes are shorter. This result shows that the simultaneous application of both outlier exclusion strategies, interquartile and percentile on both amplitude and phase series, on Test 1 eliminated more noisy values than on Test 2. Therefore, although Test 1 (99.1%) has a slightly lower accuracy than Test 2 (99.4%), Test 1 is preferable as a model as it has classification errors with smaller deviations around the right liquid volumes.

The use of distance-based machine learning algorithms was prominent in this work, as the results achieved accuracies of 99.4% for 1-NN and 97.3% for Rocchio, with both using Manhattan distance metrics. Values of 3-NN = 99.1% and 5-NN = 98.5% were calculated, proving that one-neighborhood models have better accuracy. The accuracies for Euclidean distance in sixth scenario were 88.5% for Rocchio and 98.0% for 1-NN.

Manhattan metrics represent the distance between two points measured along axes at right angles of 90° degrees, following a grid approach instead of the straight lines of Euclidean distance [67]. When comparing measured and reference signals, similarity methods based on Manhattan metrics can be used to minimize the distances between signals [68,69]. Besides the better accuracy, Manhattan metrics are more preferable than Euclidean in terms of resource consumption, as Euclidean requires square-root operations, and Manhattan needs only the absolute value of a subtraction [70].

4.7. Gaussian Process Regression

A Gaussian Process Regression (GPR) algorithm was configured with the method of rational quadratic type, and the resulting R^2 of 1.00 and the RMSE of 0.21164 are shown in Figure 23. An error of 0.2 over supervised data collected with the steps of 1.0 mL shows a great fit of the GPR over the nature of physics involved in this experiment.



Figure 23. Gaussian process regression (rational quadratic) for the FOH system.

Therefore, ending the results and discussion, different machine learning algorithms were evaluated, in which two were of classification and one was of regression type. A model fit to the unwrapped phase was also shown. The classification algorithms were assessed considering five different distance metrics.

After all analyses, the best results are presented in Table 3.

Rocchio

Machine Le	arning Algorithms	Distance/Type	Metrics	Value
	k-NN 1	Manhattan		99.4%
Classification	Rocchio	Manhattan	- Accuracy -	97.3%
Classification	k-NN 1	Euclidean		93.1%

Euclidean

75.1%

Table 3. Results of machine learning algorithms used for liquid volume predictions.

Machine Le	earning Algorithms	Distance/Type	Metrics	Value
	k-NN 1	Mahalanobis		90.6%
	Rocchio	Mahalanobis		76.1%
	k-NN 1	Cosine Distance		90.9%
	Rocchio	Cosine Distance		64.6%
Regression	Sum of Sines	NT:	RMSE%	4.15%
		Nine terms	R ²	0.9994
	Gaussian Process	Pational Quadratia	RMSE	0.21164 mL
		Kauonai Quauranc	R ²	1.00

Table 3. Cont.

5. Conclusions and Future Work

In this work, a fiber-optic hydrophone (FOH) based on Michelson's interferometer (MI) is actively stabilized by an electronic feedback (EF) loop circuit to mitigate mechanical and thermal noise coming from the external environment. The FOH system is employed as a hardware and software application to perform liquid volume measurements of a 1000 mL glass graduated cylinder filled with water with volumes ranging from 440 mL to 1000 mL. This range limitation is due to the setup and positioning of the ultrasound source inside the cylinder. The sensing elements are composed of two optical fiber coils placed at the sensor head and a third optical fiber coil placed at the MI's reference arm. The coils are made of readily available and common SMF fiber. The reference coil is wound around a piezoelectric actuator to stabilize the system at the quadrature point of $\pi/2$, and the two sensor coils are submerged in water to detect acoustic wave patterns formed by a particular liquid volume, supplying distinct values of amplitudes (volts) and phases (degrees) as output measurands. Finally, due to the nonlinear behavior of theses measurands, machine-learning algorithms are employed under supervised learning to predict the liquid volumes.

The accuracy of 99.4% achieved by this work is a prominent result in liquid volume predictions for a system that employs acoustic detection through optical fiber sensors. Further, an RMSE of 0.21164 mL shows the feasibility of the FOH system. For the test tank used, the value of 0.21 mL is equivalent to 0.0714 mm of error in the height of the water column. An added contribution of this work is the characterization of the unwrapped phase measurand, in which a sum-of-sines fitting function reached an R² of 0.9994 and an RMSE% of 4.15%.

Another contribution of this work is a practical demonstration of the homodyne demodulation approach working with an active stabilization mechanism over a Michelson's interferometer, as [40] mentions that it is a challenge to stabilize such systems for sensing applications, and it is useful for laboratory analysis. The advantages and drawbacks of quadrature stabilization methods are related to the resolution of the interferometer, which is the smallest physical quantity that a sensor can measure, defined by the noise of the measurand. Multiple reflections in optical arms improve the resolution of a homodyne interferometer but can induce phase jumps and loss of laser coherence. Therefore, a compromise must be found between increasing resolution and losing coherence [35].

As a limitation, this first version of the FOH system was used to measured only one type of liquid (water), although the two optical coils in MI's sensor head were intentionally constructed to find different fluids around them (e.g., water and oil), which will be done in future tests. Another limitation is the medium time of three seconds to collect and process one data sample. As a milliliter requires twenty samples, the FOH system takes a minute to predict liquid volume.

As suggestions, the system can be improved to detect different fluids and multifluid interfaces. This stabilized FOH can also be used with a few adaptations to perform ultrasound piezoelectric transducer calibration [18,58]. Third, the FBG structure purposed

by [23,71] could include an optical coil inside the polymer diaphragms as an additional sensing mechanism to detect vibration patterns of liquids inside a production vessel. Finally, a long short-term memory (LSTM) neural network could be considered as a prediction model, as its recursive behavior would be capable of detecting a sequence of points in the "snail paths" shown in the 2D plane of modal amplitude and phase patterns.

Author Contributions: Conceptualization, W.S.D. and C.A.R.D.; methodology, W.S.D., C.A.R.D. and A.F.; software, W.S.D.; validation, W.S.D.; formal analysis, W.S.D.; resources, W.S.D., C.A.R.D. and A.F.; data curation, W.S.D.; writing—original draft preparation, W.S.D.; writing—review and editing, W.S.D., C.A.R.D., A.G.L.-J. and A.F.; supervision, C.A.R.D. and A.F.; project administration, A.F. All authors have read and agreed to the published version of the manuscript.

Funding: This research is financed by FAPES (84336650), FAPES (459/2021), CNPq (304049/2019-0), CNPq (310668/2021-2), and Petrobras (2017/00702-6).

Data Availability Statement: Data and software are available by reasonable request.

Acknowledgments: W.S.D. thanks Moisés Renato Nunes Ribeiro for the first ideas for this work. Thanks to Rogério Ramos and Ligia Gaigher Franco from NEMOG/UFES Laboratory, who both were always supportive and offered the ultrasound piezoelectric transducer, signal generator, power amplifier, acoustic absorber, and other materials used in the experiment setup.

Conflicts of Interest: The authors declare no conflict of interest. The funders had no role in the design of the study; in the collection, analyses, or interpretation of data; in the writing of the manuscript, or in the decision to publish the results.

References

- Forestieri, E.; Secondini, M.; Poti, L.; Cavaliere, F.; Member, S.; Secondini, M.; Member, S.; Pot, L.; Member, S.; Cavaliere, F. High-Speed Optical Communications Systems for Future WDM Centralized Radio Access Networks. *J. Lightwave Technol.* 2022, 40, 368–378. [CrossRef]
- Diaz, C.A.R.; Leal-junior, A.G.; Avellar, L.M.; Antunes, P.F.C.; Pontes, M.J.; Marques, C.A.; Friezera, A.; Ribeiro, M.R.N. Perrogator: A portable energy-efficient interrogator for dynamic monitoring of wavelength-based. *Sensors* 2019, 19, 2962. [CrossRef] [PubMed]
- Cole, J.H.; Bucaro, J.A.; Kirkendall, C.K.; Dandridge, A. The origin, history and future of fiber-optic interferometric acoustic sensors for US Navy applications. In Proceedings of the 21st International Conference on Optical Fiber Sensors, Ottawa, ON, Canada, 15–19 May 2011; Volume 7753, p. 775303. [CrossRef]
- 4. Bucaro, J.A.; Dardy, H.D.; Carome, E.F. Fiber-optic hydrophone. J. Acoust. Soc. Am. 1977, 62, 1302–1304. [CrossRef]
- Campanella, C.E.; Cuccovillo, A.; Campanella, C.; Yurt, A.; Passaro, V.M.N. Fibre Bragg Grating based strain sensors: Review of technology and applications. *Sensors* 2018, 18, 3115. [CrossRef]
- Culshaw, B. Optical fibre sensors: A compressed perspective. In Proceedings of the IEEE Sensors, Scotland, UK, 29 October–1 November 2017; Institute of Electrical and Electronics Engineers Inc.: Piscataway, NJ, USA, 2017; pp. 1–3.
- Yin, S.; Ruffin, P.B.; Yu, F.T.S. Fiber Optic Sensors, 2nd ed.; Thompson, B.J., Ed.; CRC Press-Taylor & Francis Group, LLC: Boca Raton, FL, USA, 2008; ISBN 978-1-4200-5365-4.
- Leal-Junior, A.G.; Marques, C.; Frizera, A.; Pontes, M.J. Multi-interface level in oil tanks and applications of optical fiber sensors. Opt. Fiber Technol. 2018, 40, 82–92. [CrossRef]
- da Silva Marques, R.; Prado, A.R.; da Costa Antunes, P.F.; de Brito André, P.S.; Ribeiro, M.R.N.; Frizera-Neto, A.; Pontes, M.J. Corrosion resistant FBG-based quasi-distributed sensor for crude oil tank dynamic temperature profile monitoring. *Sensors* 2015, 15, 693–703. [CrossRef]
- 10. Hjertaker, B.T. Level measurement and control strategies for subsea separators. J. Electron. Imaging 2001, 10, 679–689. [CrossRef]
- 11. Hjertaker, B.T.; Johansen, G.A.; Jackson, P. Recent developments in hydrocarbon separator interface imaging. In *Process Imaging for Automatic Control*; Society of Photo Optical: Bellingham, WA, USA, 2001.
- 12. Rupitsch, S.J. Piezoelectric Sensors and Actuators: Fundamentals and Applications; Springer: Berlin/Heidelberg, Germany, 2019; ISBN 978-3-662-57532-1.
- 13. Morris, P.; Hurrell, A.; Zhang, E.; Rajagopal, S.; Beard, P. A fabry-perot fibre-optic hydrophone for the measurement of ultrasound induced temperature changes. *Proc. IEEE Ultrason. Symp.* **2006**, *1*, 536–539. [CrossRef]
- 14. De Paula, R.P.; Cole, J.H.; Bucaro, J.A. Broad-Band Ultrasonic Sensor Based on Induced Optical Phase Shifts in Single-Mode Fibers. *J. Light. Technol.* **1983**, *1*, 390–393. [CrossRef]
- 15. Meng, Z.; Chen, W.; Wang, J.; Hu, X.; Chen, M.; Zhang, Y. Recent Progress in Fiber-Optic Hydrophones. *Photonic Sens.* 2021, 11, 109–122. [CrossRef]

- Ma, J.; Zhao, M.; Huang, X.; Bae, H.; Chen, Y.; Yu, M.; Inyu, J.M.A.; Hao, M.E.Z.; Uang, X.I.H.; Ae, H.Y.B.; et al. Low cost, high performance white-light fiber-optic hydrophone system with a trackable working point. *Opt. Express* 2016, 24, 19008. [CrossRef] [PubMed]
- 17. Wurster, C.; Staudenraus, J.; Eisenmenger, W. The fiber optic probe hydrophone. *Proc. IEEE Ultrason. Symp.* **1994**, *2*, 941–944. [CrossRef]
- Chang, S.; Zhu, H.; Yang, P. An Active Stabilization Technique of A Homodyne Interferometer Based On PTDC For High Frequency Hydrophone Calibration. In Proceedings of the 2015 3rd International Conference on Mechanical Engineering and Intelligent Systems, Yinchuan, China, 15–16 August 2015; Volume 26, pp. 317–324. [CrossRef]
- Yang, Y.; Luo, Z.; Wang, Z.; Zhang, Y.; Chang, T.; Yu, M.; Cheng, L.; Wu, C.; Cui, H.L. High-Performance Fiber Optic Interferometric Hydrophone Based on Push-Pull Structure. *IEEE Trans. Instrum. Meas.* 2021, 70, 9511313. [CrossRef]
- 20. Buick, J.M.; Cosgrove, J.A.; Douissard, P.A.; Greated, C.A.; Gilabert, B. Application of the acousto-optic effect to pressure measurements in ultrasound fields in water using a laser vibrometer. *Rev. Sci. Instrum.* **2004**, *75*, 3203–3207. [CrossRef]
- 21. Fan, H.; Zhang, L.; Gao, S.; Chen, L.; Bao, X. Ultrasound sensing based on an in-fiber dual-cavity Fabry–Perot interferometer. *Opt. Lett.* **2019**, *44*, 3606–3609. [CrossRef]
- 22. Fan, H.; Chen, L.; Bao, X. Chalcogenide microfiber-assisted silica microfiber for ultrasound detection. *Opt. Lett.* **2020**, *45*, 1128. [CrossRef]
- Díaz, C.A.R.; Leal-junior, A.G.; André, P.S.B.; Da Costa Antunes, P.F.; Pontes, M.J.; Frizera-Neto, A.; Ribeiro, M.R.N. Liquid level measurement based on FBG-embedded diaphragms with temperature compensation. *IEEE Sens. J.* 2018, 18, 193–200. [CrossRef]
- Vorathin, E.; Hafizi, Z.M.; Ismail, N.; Loman, M. Review of high sensitivity fibre-optic pressure sensors for low pressure sensing. Opt. Laser Technol. 2020, 121, 105841. [CrossRef]
- 25. Wild, G.; Hinckley, S. Acousto-ultrasonic optical fiber sensors: Overview and state-of-the-art. *IEEE Sens. J.* **2008**, *8*, 1184–1193. [CrossRef]
- Yang, F.; Zhang, W.; Zhao, S.; Liu, Q.; Tao, J.; He, Z. Miniature interrogator for multiplexed FBG strain sensors based on a thermally tunable microring resonator array. *Opt. Express* 2019, 27, 6037–6046. [CrossRef]
- 27. Massey, G.A. An Optical Heterodyne Ultrasonic Image Converter. Proc. IEEE 1968, 56, 2157–2161. [CrossRef]
- 28. Moss, G.E.; Miller, L.R.; Forward, R.L. Photon-Noise-Limited Laser Transducer for Gravitational Antenna. *Appl. Opt.* **1971**, 10, 2495. [CrossRef] [PubMed]
- 29. Cole, J.H.; Johnson, R.L.; Bhuta, P.G. Fiber-optic detection of sound. J. Acoust. Soc. Am. 1977, 62, 1136–1138. [CrossRef]
- Hu, J.; Liu, D.; Liu, H. Research on noise floor of an interferometric fiber-optic hydrophone system. In Proceedings of the 2017 IEEE International Conference on Signal Processing, Communications and Computing (ICSPCC 2017), Xiamen, China, 22–25 October 2017; pp. 1–4. [CrossRef]
- 31. Krohn, D.; MacDougall, T.; Mendez, A. *Fiber Optic Sensors: Fundamentals and applications*, 4th ed.; Society of Photo-Optical Instrumentation Engineers (SPIE) All: Bellingham, WA, USA; Washington, DC, USA, 2014; ISBN 978-1-6284-1180-5.
- 32. Lee, B.H.; Kim, Y.H.; Park, K.S.; Eom, J.B.; Kim, M.J.; Rho, B.S.; Choi, H.Y. Interferometric Fiber Optic Sensors. *Sensors* 2012, 12, 2467–2486. [CrossRef] [PubMed]
- Meribout, M.; Naamany, A.A.; Busaidi, K. Al An acoustic system for providing the two-phase liquid profile in oil field storage tanks. *IEEE Trans. Ultrason. Ferroelectr. Freq. Control* 2009, 56, 2241–2250. [CrossRef]
- 34. Liang, X.; Ren, G.; Li, Y.; Liu, Z.; Wei, H.; Jian, S. In-fiber liquid-level probe based on Michelson interferometer via dual-mode elliptical multilayer-core fiber. *J. Mod. Opt.* **2016**, *63*, 1254–1259. [CrossRef]
- 35. Watchi, J.; Cooper, S.; Ding, B.; Mow-Lowry, C.M.; Collette, C. Contributed Review: A review of compact interferometers. *Rev. Sci. Instrum.* 2018, *89*, 121501. [CrossRef]
- 36. Staudenraus, J.; Eisenmenger, W. Fibre-optic probe hydrophone for ultrasonic and shock-wave measurements in water. *Ultrasonics* **1993**, *31*, 267–273. [CrossRef]
- Shen, B.; Wada, Y.; Koyama, D.; Isago, R.; Mizuno, Y.; Nakamura, K. Fiber-optic ultrasonic probe based on refractive-index modulation in water. In Proceedings of the 21st International Conference on Optical Fiber Sensors, Ottawa, ON, Canada, 15–19 May 2011; Volume 7753, p. 77539W. [CrossRef]
- Shao, L.Y.; Lau, S.T.; Dong, X.; Zhang, A.P.; Chan, H.L.W.; Tam, H.Y.; He, S. High-frequency ultrasonic hydrophone based on a cladding-etched DBR fiber laser. *IEEE Photonics Technol. Lett.* 2008, 20, 548–550. [CrossRef]
- Hu, J.; Li, D. Simulation and testing of a noise-limited demodulation system for a fiber-optic hydrophone system based on a Michelson interferometer. In Proceedings of the IEEE/OES China Ocean Acoustics Symposium (COA2016), Harbin, China, 9–11 January 2016. [CrossRef]
- 40. Kirkendall, C.K.; Dandridge, A. Overview of high performance fibre-optic sensing. J. Phys. D Appl. Phys. 2004, 37, R197. [CrossRef]
- Fan, H.; Ma, W.; Chen, L.; Bao, X. Ultracompact twisted silica taper for 20 kHz to 94 MHz ultrasound sensing. *Opt. Lett.* 2020, 45, 3889. [CrossRef] [PubMed]
- Kimura, K.; Takeuchi, S.; Koike, Y. Investigation of Cavitation Bubble Influence on Frequency Spectrum of Fiber Optic Probe Hydrophone Output. In Proceedings of the 2020 IEEE International Ultrasonics Symposium (IUS), Las Vegas, NV, USA, 7–11 September 2020.

- Davis, A.R.; Kirkendall, C.K.; Dandridge, A.; Kersey, A.D. 64 Channel All Optical Deployable Acoustic Array. In Optical Fiber Sensors; Optical Society of America: Washington, DC, USA, 2015; Volume 16, p. OFA6. [CrossRef]
- Chan, H.L.W.; Chiang, K.S.; Price, D.C.; Gardener, J.L.; Brinch, J.; Gardner, J.L.; Brinch, J.; Gardener, J.L.; Brinch, J. Use of a fibre-optic hydrophone in measuring acoustic parameters of high power hyperthermia transducers. *Phys. Med. Biol.* 1989, 34, 1609–1622. [CrossRef] [PubMed]
- 45. Morris, P.; Beard, P.; Hurrell, A. Development of a 50MHz optical fibre hydrophone for the characterisation of medical ultrasound fields. *Proc. IEEE Ultrason. Symp.* 2005, *3*, 1747–1750.
- 46. Al-Naamany, A.M.; Meribout, M.; Al Busaidi, K. Design and implementation of a new nonradioactive-based machine for detecting oil-water interfaces in oil tanks. *IEEE Trans. Instrum. Meas.* 2007, *56*, 1532–1536. [CrossRef]
- Beard, P.C.; Hurrell, A.M.; Mills, T.N. Characterization of a polymer film optical fiber hydrophone for use in the range 1 to 20 MHz: A comparison with PVDF needle and membrane hydrophones. *IEEE Trans. Ultrason. Ferroelectr. Freq. Control* 2000, 47, 256–264. [CrossRef]
- 48. Wang, M.; Triantafillos, K.; Xing, G.; He, L.; Yang, P.; Zhang, Y. Measurement of underwater acoustic pressures in the frequency range 25 to 500 kHz using optical interferometry and discussion on associated uncertainties. In Proceedings of the 25th International Congress on Sound and Vibration 2018 (ICSV 25), Hiroshima, Japan, 8–12 July 2018; Volume 8, pp. 4909–4914.
- Wang, W.; Pei, Y.; Ye, L.; Song, K. High-Sensitivity Cuboid Interferometric Fiber-Optic Hydrophone Based on Planar Rectangular Film Sensing. Sensors 2020, 20, 6422. [CrossRef] [PubMed]
- 50. Faisal, S.; Bukhari, A.; Yang, W. Multi-interface Level Sensors and New Development in Monitoring and Control of Oil Separators. Sensors 2006, 6, 380–389. [CrossRef]
- 51. Wang, C.C.; Dandridge, A.D.; Tveten, A.B.; Yurek, A.M.; Wang, C.C.; Dandridge, A.D.; Tveten, A.B.; Yurek, A.M. Very high responsivity fiber optic hydrophones for commercial applications. *SPIE* **1994**, *2360*, *360*–*363*. [CrossRef]
- 52. Udd, E.; William, B.; Spillman, J. Fiber Optic Sensors: An Introduction for Engineers and Scientists, 2nd ed.; Wiley: Hoboken, NJ, USA, 2011; ISBN 978.
- 53. Reider, G.A. *Photonics: An Introduction;* Springer International Publishing: Cham, Switzerland; Vienna, Austria, 2016; ISBN 978-3-319-26074-7.
- 54. Grassani, D.; Galli, M.; Bajoni, D. Active stabilization of a Michelson interferometer at an arbitrary phase with subnanometer resolution. *Opt. Lett.* **2014**, *39*, 2530. [CrossRef]
- 55. Xie, F.; Chen, Z.; Ren, J. Stabilisation of an optical fiber Michelson interferometer measurement system using a simple feedback circuit. *Meas. J. Int. Meas. Confed.* **2009**, *42*, 1335–1340. [CrossRef]
- 56. Elezov, M.S.; Scherbatenko, M.L.; Sych, D.V.; Goltsman, G.N. Active and passive phase stabilization for the all-fiber Michelson interferometer. *J. Phys. Conf. Ser.* 2018, 1124, 051014. [CrossRef]
- 57. Bucaro, J.A.; Dardy, H.D.; Carome, E.F. Optical fiber acoustic sensor. Appl. Opt. 1977, 16, 1761–1762. [CrossRef] [PubMed]
- 58. Preston, R.C.; Robinson, S.P.; Zeqiri, B.; Esward, T.J.; Gélat, P.N.; Lee, N.D. Primary calibration of membrane hydrophones in the frequency range 0.5 MHz to 60 MHz. *Metrologia* **1999**, *36*, 331. [CrossRef]
- 59. MATLAB, Version 9.11.0.1873467. (R2021b) Update 3 (Student License). The Math Works Inc.: Natick, MA, USA, 2021.
- 60. Lubbers, J.; Graaff, R. A simple and accurate formula for the sound velocity in water. *Ultrasound Med. Biol.* **1998**, 24, 1065–1068. [CrossRef]
- 61. Jarzynski, J.; Hughes, R. Static pressure sensitivity amplification in interferometric fiber-optic hydrophones. *Appl. Opt.* **1980**, *19*, 98–107. [CrossRef]
- 62. Bucaro, J.A.; Hickman, T.R. Measurement of sensitivity of optical fibers for acoustic detection. *Appl. Opt.* **1979**, *18*, 938–940. [CrossRef] [PubMed]
- 63. Monsay, E.H.; Gilbert, D.E. Predicted performance of a heterodyne detector with a fiber optic coil hydrophone. *Ocean. Conf. Rec.* **1981**, *1*, 31–34. [CrossRef]
- 64. Wang, L.; Zhang, M.; Mao, X.; Liao, Y. The arctangent approach of digital PGC demodulation for optic interferometric sensors. *Interferom. XIII Tech. Anal.* 2006, 6292, 62921E. [CrossRef]
- 65. Sattar, F.; Driessen, P.F.; Tzanetakis, G.; Page, W.H. A new event detection method for noisy hydrophone data. *Appl. Acoust.* **2020**, 159, 107056. [CrossRef]
- 66. Rogers, S.; Girolami, M. A First Course in Machine Learning, 2nd ed.; CRC Press-Taylor & Francis Group, LLC: Boca Raton, FL, USA, 2017; ISBN 9781498738569.
- 67. Singh, H. Practical Machine Learning and Image Processing; Apress: Allahabad, India, 2019; ISBN 9781484241486.
- 68. Hartog, A.H. An Introduction to Distributed Optical Fibre Sensors; CRC Press-Taylor & Francis Group, LLC: Boca Raton, FL, USA, 2016; ISBN 978-1-4822-5957-5.
- 69. Wang, F.; Zhan, W.; Lu, Y.; Yan, Z.; Zhang, X. Determining the Change of Brillouin Frequency Shift by Using the Similarity Matching Method. *J. Lightwave Technol.* 2015, 33, 4101–4108. [CrossRef]
- 70. Saini, H.S.; Singh, R.K.; Kumar, G.; Rather, G.M.; Santhi, K. *Innovations in Electronics and Communication Engineering*; Kacprzyk, J., Ed.; Springer: Warsaw, Poland, 2018; Volume 65, ISBN 978-981-13-3764-2.
- 71. Leal-Junior, A.G.; Díaz, C.A.R.; Frizera, A.; Marques, C.; Ribeiro, M.R.N.; Pontes, M.J. Simultaneous measurement of pressure and temperature with a single FBG embedded in a polymer diaphragm. *Opt. Laser Technol.* **2019**, *112*, 77–84. [CrossRef]